Author's response to reviews

Title: The effects of multi-domain versus single-domain cognitive training in non-demented older people: a randomized controlled trial

Authors:

Yan Cheng (yancheng_research@hotmail.com)
Wenyuan Wu (wenyuanwu_tj@hotmail.com)
Wei Feng (ffww06@163.com)
Jiaqi Wang (jiamian316@163.com)
You Chen (huaganyan@163.com)
Yuan Shen (shenyuanmm@yahoo.com.cn)
Qingwei Li (lianocd@yahoo.com.cn)
Xu Zhang (zzz1082@163.com)
Chunbo Li (chunbo_li@yahoo.com)

Version: 3 Date: 9 February 2012

Author's response to reviews: see over
Dear Dr. Tree-Booker,

We would like to express our gratitude for the comments made by you and reviewers (MS: 5651524986181287). We have modified the manuscript according to the editorial revisions. All the changes we have made are typed in red. We respond to the revisions as follows.

Sincerely yours,

Wenyuan Wu

**Question 1:**

Authors' contributions: please distinguish between YC and YC. Please also confirm the role that YS, QL and XZ had in this study; their information is missing from the Authors' contributions section. Please note that to qualify as an author one should 1) have made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data; 2) have been involved in drafting the manuscript or revising it critically for important intellectual content; and 3) have given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content. Acquisition of funding, collection of data, or general supervision of the research group, alone, does not justify authorship.

**Reply:** We really appreciated your review. We added the related content in the Authors' contributions section. Thank you very much.

**Question 2:**
Please ensure that your manuscript fully adheres to the CONSORT reporting guidelines (http://www.consort-statement.org/consort-statement/overview0/#checklist)

**Reply:** Thank you for the suggestion. We have checked and revised the manuscript according to the guidelines. Thank you very much.

**Question 3:**

Please ensure that your revised manuscript conforms to the journal style (http://www.biomedcentral.com/info/ifora/medicine_journals). It is important that your files are correctly formatted.

**Reply:** We are very grateful for your comments. We have revised the manuscript according to the correct format. Thank you very much.